Author's response to reviews

Title: Efficacy of hypnosis/guided imagery in fibromyalgia syndrome - a systematic review and meta-analysis of controlled trials

Authors:

Kathrin Bernardy (kathrin.bernardy@uniklinikum-saarland.de)
Nicole Füber (nicolefueber@googlemail.com)
Petra Klose (p.klose@kliniken-essen-mitte.de)
Winfried Häuser (whaeuser@klinikum-saarbruecken.de)

Version: 5 Date: 15 June 2011

Author's response to reviews:

The following changes had been made:

Abstract: keywords removed
Structure (Text): 'Methods' made a heading.
References: functioning link to http://www.cochrane.org/ provided#
Table 1: visible vertical lines from table 1 removed
typographical errors: corrected

I was not able to crop the figures as closely as possible to minimise white space around the image. I do not know how to perform it. This cropping had not been required by any other journal to which I have submitted. Please help